Author's response to reviews

Title: Intragenic duplication in the PHKD1 gene in autosomal recessive polycystic kidney disease

Authors:

Jun Miyazaki (mjun@fujita-hu.ac.jp)
Mayuko Ito (mayuko0718jp@nifty.com)
Haruki Nishizawa (nharuki@fujita-hu.ac.jp)
Takema Kato (takema@fujita-hu.ac.jp)
Yukito Minami (yukito@fujita-hu.ac.jp)
Hidehito Inagaki (hinagaki@fujita-hu.ac.jp)
Tamae Ohye (otamae@fujita-hu.ac.jp)
Masafumi Miyata (m-miata@j4.so-net.ne.jp)
Hiroko Boda (mizujin@fujita-hu.ac.jp)
Yuka Kiriyama (ykiri@fujita-hu.ac.jp)
Makoto Kuroda (mkuro@fujita-hu.ac.jp)
Takao Sekiya (kurnsen@zg7.so-net.ne.jp)
Hiroki Kurahashi (kura@fujita-hu.ac.jp)
Takuma Fujii (fujit44@gmail.com)

Version: 8 Date: 7 October 2015

Author's response to reviews: see over
Dear Editor,

We are submitting our revised manuscript entitled, “Intragenic duplication in the PHKD1 gene in autosomal recessive polycystic kidney disease” to be considered for publication in *BMC Medical Genetics*. We have revised the manuscript according to the reviewers’ suggestions. We hope that we have sufficiently addressed all the points of concern raised by the reviewers and that the manuscript has been improved to the level for publication in your journal.

If there are any questions about this manuscript, you may feel free to contact me by e-mail. Thank you very much for your consideration. I look forward to hearing from you soon.

Sincerely,

Hiroki Kurahashi, MD, PhD
Division of Molecular Genetics
Institute for Comprehensive Medical Science, Fujita Health University
Toyoake, Aichi 470-1192, Japan
Phone: +81-562-93-9391
FAX: +81-562-93-8831
e-mail: kura@fujita-hu.ac.jp